Author's response to reviews

Title: Clearance of Technetium-99m-DTPA and HRCT Findings in the Evaluation of Patients with Idiopathic Pulmonary Fibrosis

Authors:

Katerina M Antoniou (katerinaantoniou@yahoo.gr)
Katerina Malagari (kmalag@acn.gr)
Nikolaos Tzanakis (tzanakis@med.uoc.gr)
Konstantinos Perisinakis (perisina@med.uoc.gr)
Emmanouil K Symvoulakis (symvouman@yahoo.com)
Nikolaos Karkavitsas (karkavitsas@med.uoc.gr)
Nikolaos M Siafakas (siafak@med.uoc.gr)
Demosthenes Bouros (bouros@med.duth.gr)

Version: 6 Date: 9 January 2006

Author's response to reviews:

Dear Editor,

We have included the statement about the ethical approval in the end of "methods" section in our revised manuscript.

We state that it was not necessary for our project to be approved from the ethical comittee of the hospital, because all the investigations were incorporated to the routine diagnosis evaluation of IPF patients.

on behalf of the authors,

Dr Katerina Antoniou. MD, PhD